Author's response to reviews

Title: SNAI1 Expression and the Mesenchymal Phenotype: An Immunohistochemical Study performed on 46 Cases of Oral Squamous Cell Carcinoma

Authors:

Joerg Schwock (jschwock@uhnres.utoronto.ca)
Grace Bradley (bradley@uhnres.utoronto.ca)
James C Ho (jamesho@uhnres.utoronto.ca)
Bayardo Perez-Ordenez (bayardo.perez-ordonez@uhn.on.ca)
David W Hedley (david.hedley@uhn.on.ca)
Jonathan C Irish (jonathan.irish@uhn.on.ca)
William R Geddie (william.geddie@uhn.on.ca)

Version: 3 Date: 13 January 2010

Author's response to reviews: see over
Dear Dr. Burgess,

Thank you for considering our manuscript for publication in *BMC Clinical Pathology*!

As requested in your e-mail dated January 12th-2010, we have amended the Material and Methods section with a statement which follows our declaration of ethical approval of the research study as given in our Author’s Response letter provided with the previous submission.

We would like to thank you for your support and guidance through the review process!

With kind regards,

Joerg Schwock Dr.med. (GER)
William R Geddie MD FRCPC

Toronto, January 13 - 2010